Author's response to reviews

Title: Feeding mice with diets containing mercury-contaminated fish flesh from French Guiana: a model for the mercurial intoxication of the Wayana Amerindians

Authors:

Jean-Paul Bourdineaud (jp.bourdineaud@epoc.u-bordeaux1.fr)
Nadege Bellance (nadege.bellance@etud.u-bordeaux2.fr)
Giovani Benard (benardg@umbi.umd.edu)
Daniel Brethes (daniel.brethes@lbgc.u-bordeaux2.fr)
Masatake Fujimura (fujimura@nimd.go.jp)
Patrice Gonzalez (p.gonzalez@epoc.u-bordeaux1.fr)
Aline Marighetto (a.marighetto@cnic.u-bordeaux1.fr)
Regine Maury-Brachet (r.maury-brachet@epoc.u-bordeaux1.fr)
Cecile Mormede (c.mormede@cnic.u-bordeaux1.fr)
Vanessa Pedron (pedron_vanessa@yahoo.fr)
Jean-Nicolas Philippin (jn.philippin@cnic.u-bordeaux1.fr)
Rodrigue Rossignol (rossig@u-bordeaux2.fr)
William Rostene (rostene@st-antoine.inserm.fr)
Masumi Sawada (sawada@nimd.go.jp)
Muriel Laclau (m.laclau@epoc.u-bordeaux1.fr)

Version: 5 Date: 10 October 2008

Author's response to reviews: see over
Cover letter

To the editorial production team,

Please find the revised version of our article.
We took into consideration all the necessary changes:
- the references 45-59 are now cited within the manuscript text.
- table 5 is now cited.
- all the tables have been created now using Tables tool from Microsoft.
- about the ethical approval when dealing with mice experiments: the corresponding statement was already appearing in the text; one can find it in the Material & Methods section under the paragraph entitled “Mice treatment and tissue sampling”, lines 4-5.
- the page numbers have been removed.
- the statement indicating that all authors read and approved the final manuscript is now appearing at the end of the “Authors’ contribution” paragraph.

Yours sincerely
Jean-Paul Bourdineaud